Author's response to reviews

Title: Investigation of the expression of the EphB4 receptor tyrosine kinase in prostate carcinoma

Authors:

Yen-Ching Lee (ken.lee@student.adelaide.edu.au)
Janeanne R Perren (janeanne.perren@student.adelaide.edu.au)
Evelyn L Douglas (evelyn.douglas@adelaide.edu.au)
Michael P Raynor (mraynor@primus.com.au)
Maria A Bartley (ants75@hotmail.com)
Peter G Bardy (Peter.Bardy@imvs.sa.gov.au)
Sally-Anne Stephenson (sally-anne.stephenson@adelaide.edu.au)

Version: 3 Date: 2 September 2005

Author's response to reviews:

Thank you for accepting our paper for publication in BMC Cancer. As suggested by Reviewer Sugimura we have indicated that this is a pilot study, not by adding a subtitle, but by altering the final sentence of the background section. In regard to the Discretionary revision suggested by Reviewer Kinch we have decided not to further alter the manuscript although we acknowledge the comment.

Sincerely,

Sally Stephenson